Author's response to reviews

Title: Integrating the promotion of physical activity within a smoking cessation programme: Findings from collaborative action research in UK stop smoking services.

Authors:

Adrian H Taylor (A.H.Taylor@exeter.ac.uk)
Emma S Everson-Hock (e.s.everson-hock@shef.ac.uk)
Michael Ussher (mussher@sgul.ac.uk)

Version: 2 Date: 7 July 2010

Author's response to reviews: see over
Dear Editor,

Pleased find enclosed a revised manuscript to be considered for BMC Health Services Research entitled: **Integrating the promotion of physical activity within a smoking cessation programme: Findings from collaborative action research in UK stop smoking services.**

We thank the reviewers and editor for their critical and constructive comments which we feel have strengthened the manuscript considerably. All the authors have been personally and actively involved in substantive work leading to the paper and the revisions, and will hold themselves jointly and individually responsible for its content. I have assumed responsibility for keeping my co-authors informed of progress through the editorial review process.

We look forward to receiving the editor’s and reviewers' feedback.

Best wishes,

Prof Adrian Taylor

(Professor of Exercise & Health Psychology)